Author's response to reviews

Title: Healthcare decision-making in end stage renal disease-patient preferences and clinical correlates

Authors:

Anuradha Jayanti (anu_5577@yahoo.co.in)
Markus Neuvonen (Markus.Neuvonen@filosofianakatemia.fi)
Alison Wearden (Alison.Wearden@manchester.ac.uk)
Philip Foden (Philip.Foden@manchester.ac.uk)
Julie Morris (Julie.Morris@manchester.ac.uk)
Paul Brenchley (Paul.Brenchley@manchester.ac.uk)
Sandip Mitra (sandip.mitra@cmft.nhs.uk)

Version: 2
Date: 21 March 2015

Author's response to reviews:

1. Line Numbering and Page numbers provided
2. Title Page now includes email addresses of all authors
3. Introduction renamed as Background
4. Conclusions provided separately, with study limitations subsection. Introduction to discussion identified key strengths, hence not repeated.
5. Consent updated under study registration in Methods
6. Abbreviations provided after acknowledgements
7. Disclosures changed to Competing Interests section
8. Author's contributions added ahead of Acknowledgements
9. Figure Legends provided after References section